

1

2



Dipartimento di Economia Politica



Materiali di discussione

\\ 655 \\

Exports and Italy's Economic Development: a Long-Run Perspective (1863-2004)

> Barbara Pistoresi¹ Alberto Rinaldi²

> > May 2011

Università di Modena e Reggio Emilia Dipartimento di Economia Politica and ReCent Via Berengario, 51 41100 Modena, Italy e-mail: <u>paola.bertolini@unimore.it</u> e-mail: <u>alberto.rinaldi@unimore.it</u>

ISSN: 2039-1439 a stampa **ISSN: 2039-1447** on line



Exports and Italy's economic development: a long-run perspective (1863-2004)

Barbara Pistoresi and Alberto Rinaldi^{*}

Dipartimento di Economia Politica and ReCent, Università di Modena e Reggio Emilia, Viale Berengario 51, 41100 Modena, Italy

Abstract

This paper investigates the relationship between real export and real GDP in Italy from 1863 to 2004 by using cointegration analysis and causality tests. The outcome suggests that these variables comove in the long run but the direction of causality depends on the level of economic development: in the period prior to WW1 the growth of the Italian economy led that of exports, while in the post-WW2 period the causal relationship was reversed with the expansion of exports that determined the growth of the Italian economy.

JEL Code: F43; O11; N1; N7

Keywords: Export led growth hypothesis, unit root tests, cointegration analysis, Granger - causality.

^{*} Corresponding author. E-mail: <u>alberto.rinaldi@unimore.it</u>. The authors would like to thank Giovanni Federico and Michelangelo Vasta for letting them have access to the new official series of Italian foreign trade of the Bank of Italy before they were published.

1. Introduction

The focus on the foreign exchange constraint in economic development and the role of exports as a determinant of economic growth owes much to the early contributions by Nurkse (1961), McKinnon (1964), Keesing (1967) and Kaldor (1970).

There are a number of reasons why exports should lead economic growth. Firstly, export growth directly increases the aggregate demand and then real output; moreover, by loosening the foreign exchange constraint, it makes easier to import inputs to meet domestic demand, and so enables output expansion (McKinnon 1964; Chenery and Strout 1966; Balassa 1978; Esfahani 1991).

Secondly, an expansion in exports may promote the reallocation of resources from a relatively low productivity non-export sector to a high productivity export sector. Higher productivity may in turn lead to output growth (Verdoorn 1949).

Lastly, export growth may promote the diffusion of technical knowledge (Grossman and Helpman 1991) and enhance efficiency through the international competition (Krueger, 1980). It may allow the exploitation of economies of scale if domestic markets are too small for optimal scale. All these factors may lead to higher economic growth.

However, the support for the export led growth (ELG) hypothesis is not universal and some strands of literature advocate for *growth-led exports* (GLE) or *feedback relationship* between exports and output.

For example, neoclassical trade theory states that economic growth leads to enhancement of skills and technology, with this increased efficiency creating a comparative advantage for the country that facilitates exports (Lancaster 1980; Krugman 1984). Moreover, exports may rise from the realization of economies of scale due to productivity gains; the rise in exports may in turn enable further cost reductions, which may result in further productivity gains and output increase (Helpman and Krugman 1985).

Finally, there is potential for no causal relationship between exports and economic growth when the growth paths of the two variables are determined by other unrelated variables (i.e., investment) (Pack 1988).

These different arguments on the exports-growth nexus suggested by the development and growth literature generated abundant empirical studies. However, these studies have not provided a conclusive response on the direction of causality between export and economic growth.

One reason for this lack of univocity in the results is due to the different econometric techniques that have been employed. Giles and Williams (2000a) presented a comprehensive survey of more than 150 applied papers on the relationship between exports and economic growth distinguishing between cross sections and time series approach. In general, cross section analyses find support for the ELG hypothesis, but this result is not confirmed by time series studies.¹

The early studies consist mostly of OLS linear models (both simple and cross country regressions) in which a growth variable is regressed on an export variable. The ELG hypothesis is supported if the coefficient on the export variable is positive and statistically significant.

These regressions provide little insight into the way the various right-hand side variables affect growth and the dynamic behaviour within countries; given the possible simultaneity involved in such models the positive association is as compatible with the reverse causation in which growth promotes export (GLE hypothesis) as with ELG or feedback effects.

In addition, these models have implicitly assumed that the regression parameters are constant across countries and that the variables involved are statistically stationary. Finally, they estimates some short run dynamics between exports and growth but do not permit the estimation of long run equilibrium states.

A more recent strand of literature applies various time series techniques to examine the exportsgrowth nexus and avoid these potential problems with the cross-section methods. These include the

¹ See Giles and Williams (2000 b) for a discussion of the robustness of econometric methods used in these studies.

time series properties of the data used, the analysis of the long run comovement of the variables (i.e cointegration analysis), tests for causation and stability of the estimated relationships.

Another reason for the unconclusiveness of the results on ELG is due to the arbitrary division in many studies of the countries surveyed in relation to their level of development (data selection bias). This aspect concerns the existence of a difference in the effect of exports on economic growth between countries above and below some critical level of some variable, indicated as 'threshold effect'.

Several authors (Michaely 1977; Tyler 1981; Kavoussi 1984; Ram 1987; Greenaway and Sapsford 1991; Burney 1996) find that the export-growth nexus tended to be stronger in higher income economies than in low income ones. Yaghmaian and Ghorashi (1995) explain the 'threshold effect' by stressing that industrialization in developing countries begins with the gradual erosion of precapitalist modes of production. The nascent industrialization leads to an increase in total factor productivity by shifting resources from low-productivity sectors to higher productivity ones. It is only after having gained the competitive edge and cost advantage in certain products that the more competitive firms in developing countries enter the world market in search of demand for their products. In this scenario, ELG is preceded by economic development and structural change in the economy.

However, Balassa (1984) came to different results and found that, for a given increment in exports, economic growth is higher the lower is the level of development. Moschos (1989) criticized earlier studies for imposing an arbitrary division between more and less advanced developing countries. He searched for a critical switching point – which can be best found from the data themselves rather than from reference to some arbitrary criterion – below and above which the responses of output growth to its determining factors differ substantially. His results too contradict the view that among high income developing economies the effect of export expansion on economic growth is stronger than among low income ones.

A large part of the literature on ELG hypothesis analyse the casual link between export and economic growth in developing nations and in particular in East Asian newly industrializing countries. Studies on Western industrialized countries are scant (Sharma *et. al.* 1991; Ghartey 1993; Awokuse 2003, 2006; Dar and Amirkhalkhali 2003) and only a few of them deal with Italy.

In particular, there are not long run interpretations or econometric tests of the relationship between exports and Italy's economic growth from Unification in 1861 to present days. The various analyses proposed are usually concerned with single and relatively short phases of Italy's economic history and consist mainly of qualitative studies.

This paper contributes to fill this gap by investigating the causal relationship between real exports and real GDP in Italy from 1863 to 2004 by using the new official series of Italian foreign trade of the Bank of Italy.

Italy provides a classic case study for the ELG hypothesis because it is a late-comer which catched up with industrialization in the late XIX century and exhibited an excellent economic performance in the second half of the XX century that enabled it to join the G-7 group.

Hence, by adopting a time series approach we test if Italian real exports and real GDP are cointegrated, then if unidirectional or bidirectional Granger causality exists, in other words if the ELG hypothesis is supported by available data for Italy.

Our results suggest that the ELG hypothesis is confirmed only for the period 1951-2004. Conversely, in the period prior to WW1 there is evidence for GLE, that is the growth of GDP led that of exports, while in the years from 1914 to 1939 we found no long run relationship between export and economic growth. Thus, our evidence is coherent with the 'threshold effect' thesis, that is, the ELG mechanism in Italy was kicked into operation only once the economy had surpassed some level of development.

This paper is organised as follows. After this Introduction, section 2 presents a review of the literature on the role of exports in Italian economic history from Unification to the beginning of the XXI century. Section 3 presents the source ad data that we have used in our analysis. Section 4

estimates an econometric model to study the relationship between real exports and real GDP in Italy from 1863 to 2004 which makes use of cointegration analysis and causality tests. Section 5 presents a historical profile of Italian exports that helps interpret the evidence of the econometric analysis. Finally, Section 6 concludes.

1. The role of exports in the debate on Italy's economic growth

There are not long run interpretations or econometric tests of the relationship between exports and Italy's economic growth from Unification to present days. The various analyses proposed are usually concerned with single and relatively short phases of Italy's economic development and consist mainly of qualitative studies. Therefore, we present a review of the economic and historical literature on the role of exports in Italy's economic growth which is subdivided in four subsections: 1) From Unification to WW1 (1861-1914); 2) The interwar period (1919-1939); 3) The 'Golden Age' (1950-1973); 4) The last thirty years (1974-2004). Lastly, a fifth subsection surveys the econometric literature that in recent years tested the ELG hypothesis for Italy for various spans of the post-WW2 period.

2.1. From Unification to WW1 (1861-1914)

A first bulk of literature focused on the fifty years after Unification. The role of trade policies – and in particular the 1887 tariff – was the most debated topic. Large part of historiography has not criticized protectionism itself, because it was seen useful to reduce the deficit of the trade balance. The criticism, starting from the seminal work of Gerschenkron (1962), regarded the sectors chosen for protection. It was assumed that the Italian government selected the wrong activities to protect: steel, textiles and wheat instead of mechanical engineering and chemicals. Fenoaltea (1973, 2006) maintained a negative view on the tariff imposed. In particular, the tariff on wheat restricted export flows, while duties on steel prevented the development of the mechanical engineering industry, which could have become a 'real' export-led sector for the Italian economy. However, this hypothesis seems too optimistic, because a nation could not easily change its specialization which is strictly linked to its technical capabilities, which are not simple input available on the market (Vasta 1999).

One of the main interpretations of that period is the so-called Bonelli-Cafagna model, which traced back some explanations of Italy's economic growth to the beginning of the XIX century. According to Bonelli (1978) and Cafagna (1989), a long wave of growth and accumulation began early in the XIX century well before Unification, stimulated by an expansion of agricultural exports, particularly raw silk. The upswing also permitted imports of raw materials and semi-manufactured goods to increase without putting pressure on the balance of payments and exchange rates. However, the agrarian crisis of the 1880s effectively ended the key role for agriculture as Italy's export engine but, by that time, other sectors had taken up the slack and a mix of emigrant remittances and tourism helped to finance growth-induced imports.

For these two authors, because Italy was a relatively backward and resource-poor country, industrialization required massive investment in plant, equipment and modern infrastructure. Since the import content of domestic output growth was large, the problem was to find the way to ease balance of payments pressures. In this context, periods of rapid economic growth among Italy's trading partners provided an impetus for domestic expansion.

The Bonelli-Cafagna model is supported by Thornton's (1997) econometric study. This author investigates the link between export and economic growth using data on real exports and real GDP for six European countries – including Italy – from the mid-XIX century to the eve of WW1. Data for the study are taken from the compilation of European historical statistics in Mitchell (1975). The approach applied is test of cointegration as a pre-test strategy for Granger tests of causality between the two variables, while trying to control for trade liberalization proxied by the ratio of total

government revenue from import duties to total imports. The results show that, for Italy, developments in real GDP and real exports were cointegrated sharing a common trend in the long run, and indicate unidirectional Granger-causality from real exports to real GDP. Thus, according to this study, the expansion of exports was a leading contributor to Italy's economic growth in that period.

However, according to Fenoaltea (1988a, 2006), who provided new estimates of Italy's investment and GDP from the Unification to WW1, the Italian economic growth must be seen in another perspective. This author explains growth and fluctuations in Italy by establishing a link between the Italian investment cycle (and economic growth) and parallel movements overseas. He argued that swings in Italian construction and in other activities were strictly supply-induced, driven by fluctuations in British capital exports that were, in turn, determined by investor sentiment in Britain. Fenoaltea proposed a financial business cycle model in which domestic economic activity was linked to international capital flows. His point was that Italy's investment cycle was essentially part of that of the financial periphery of the world economy.

So, if in the Bonelli-Cafagna view Italy's balance of payments and the international value of the lira were determined by trade flows which functioned as a constraint to economic growth, Fenoaltea argued that Itay's balance of payments and exchange rate were a function of international capital flows. As a consequence, he held that the usual contention that Italy's growth rate was constraint by balance of payments consideration is invalid. It does imply that Italy's growth depended at least in part on decisions of international investors over whom Italy had relatively little control.

This result seems in line with the econometric study by Crivellini (1993), who shows that after 1861 the contribution of exports to GDP growth was modest. However, his results remain highly tentative. There is no explicit testing of alternative hypotheses, and the econometric methods imply very strong assumptions about the exogeneity of the explanatory variables. For instance, Crivellini goes as far as assuming that not only exports but also investments and agricultural output were exogenous (Federico 1996).

2.2. The interwar period (1919-1939)

The debate on the role of exports for Italy's economic growth in *the interwar period* is not so rich. After a short recovery in the 1920s, international trade collapsed in the 1930s as a consequence of the Great Depression and return to protectionism all over in the world. Thus the role of exports was not emphasized in the analysis of Italy's economic development at that time.

2.3. The 'Golden Age' (1950-1973)

Conversely, the debate on the role of exports as a determinant of Italy's economic growth is particularly rich with regard to the years of the 'Golden Age' (1950-1973). Some authors (Stern 1967; Graziani 1969, 1998) maintained that the growth of the Italian economy was export-led at that time, although there is less agreement on exactly how exports drove the growth process. In one version, an increase in international demand for goods in which Italy had a comparative advantage induced producers to increase investment in plant and equipment and to expand output. As e result of the enlarged market, they achieved scale economies and became competitive internationally. With slack in the economy, at least initially, it may have been possible to increase output with no increase in costs or prices.

In particular, Graziani (1969, 1998) argued that for a developing nation poor of raw materials and of modern production technology such as Italy, the capacity to import was the major constraint to growth. Thus, Italy needed to boost exports to ease balance of payments pressures. On that purpose, Italy had to acquire a competitive advantage in those industries for which international demand was particularly fast-growing. That was the case of manufactured products and in particular of durables, such as furniture, cars, and electric appliances. The growth of the export sector in turn led to industrial dualism as the export sector – that used capital-intensive technology, had higher

productivity and paid higher wages – met the demand of foreign consumers whose per capita income was much higher than that of Italian ones. The latter could not afford the durables that were manufactured for exports and still demanded mostly primary goods, such as housing and foodstuffs, which were supplied by local producers that used labour-intensive technology, had lower productivity and paid lower wages.

However, other scholars have rejected – or at least reconsidered – the ELG hypothesis by maintaining that the internal demand was the driving force of the economic growth at least in the first years of the Golden Age (Ciocca, Filosa and Rey 1975; Zamagni 1992; Battilani and Fauri 2008).

In particular, Ciocca, Filosa and Rey (1975) argued that, among other things, the timing of Graziani's argument is wrong. The rapid expansion of exports post-dated the first phase of the high-growth period by almost a decade. Even among those products that experienced very fast export growth for much of the period, such as textiles, rubber goods and transportation equipment, the share of exports in total sales increased only for transport equipment. They argue further that Italy did not face a balance of payments constraint, since the growth of international trade was matched by the growth of internal demand for domestically produced goods and service. These authors maintain that government demand was the engine of growth between 1950 and 1957.

The situation changed *after 1958*, with the creation of the EEC. Trade liberalization accelerated, export growth exceeded even the blistering pace set by GDP expansion, merchandize exports outstripped tourism and emigrant remittances for the first time, and investment in machinery and equipment surpassed capital accumulation in agriculture and construction (Ciocca, Filosa and Rey 1975).

Zamagni (1992) stressed that exports expanded not only for new products such as durables but also for traditional ones such as clothing and garment and, more generally, that the role of exports in the growth of Italian industry in the years of the Golden Age must not be overplayed.

2.4. The last thirty years (1974-2004)

By contrast, the literature on the last three decades mainly focused on the structure of Italian trade and, in recent years, is becoming strictly linked to the debate on the so-called Italian 'economic decline' in the beginning of the XXI century (Gallino 2003; Visco *et al.* 2004; Vasta 2010). The rapid decrease of Italy's share on world exports represents, according to this strand of literature, one element of this decline. At the same time, the specialization of Italian exports appears too much oriented towards low-technology traditional goods. Various analyses have shown that Italian specialization is both highly concentrated in sectors with slower rate of growth in the world markets and has become more similar to that of newly developing countries (Onida 2004; De Nardis and Traù 2005). However, other studies give a less negative interpretation of the structure of Italian exports over the last 30 years. Although Italy's export specialization is largely in traditional goods, the country was able to resist competition from developing countries by shifting towards higher quality goods within the same sectors of specialization (Lanza and Quinteri 2007).

2.5. The econometric tests on the ELG hypothesis for Italy in the post-WW2 period

The first econometric studies that tested the ELG hypothesis for Italy in the post-WW2 period appeared only in the 1990s, with contrasting results.

Sharma *et al.* (1991) investigated causal relationship between real GDP growth, exports and factor inputs (capital and labour) in five industrialized countries, among which Italy. They used quarterly data for the period 1960-87 to analyse a four-variable vector autoregressive (VAR) model for each country. No casual relationship between export and GDP growth was observed for Italy, while capital was prima facie the only variable that caused growth of output.

A similar result was obtained by Pomponio (1996) who examined both bivariate causal relationship between nominal manufactured export and manufactured output growth and trivariate causal relationship between nominal manufactured exports, investment, and manufactured output for 66

OECD and less developed countries – including Italy – in the period from 1965 to 1985. Annual data were used. Before the causality testing, integration and cointegration processes were tested in order to select the appropriate functional form. The causality testing process employed a VAR approach. Both the bivariate and the trivariate analysis found no causal relationship between exports, output and investment for Italy.

Riezman *et al.* (1996) performed bivariate and trivariate Granger causality tests on the 126 countries – among which Italy – included in the purchasing power index dataset of Summers and Hesston (1991) for the years from 1950 to 1990. These authors hold that failure to account for the role of import growth can produce misleading results in the analysis of the relationship between export and income growth. They presented two alternative methods of measuring the export-GDP relationship, which allow to control for the effect of imports: the forecast error variance decomposition (FEVD) and the measure of conditional linear feedback developed by Geweke (1984). As far as Italy is concerned, bivariate results showed no causality between export and GDP growth, while trivariate results supported the export-led growth hypothesis.

A recent work by Federici and Marconi (2002) tests the ELG hypothesis for the Italian economy from 1960 to 1998 by using cointegration analysis. The authors develop a VAR model with four macroeconomic variables: an index of the OECD countries' GDP; the Italian lira's real exchange rate; Italy's quarterly data on real exports and real GDP. Their results provide clear support for the hypothesis. In particular, exports are a significant determinant of both short-run and long-run fluctuations of the Italian economy.

2. Sources and data

In 1957, Italy's Central Statistics Institute (Istat) published annual estimates of the country's historical national accounts for the period 1861-1956. These estimates included a detailed reconstruction of both production side and expenditure side at current prices, and of the latter alone at constant (1938) prices; 1938-price product series were also provided for core agriculture (cultivation and herding) and for manufacturing industry (Istat 1957). However, this work lacked key series (such as output by sector at constant prices), details on methodology and sources, and an appropriate degree of scepticism about official statistical sources (Cohen and Federico 2001).

A first attempt to improve on the Istat estimates was made a decade later by a team of scholars led by the economist Giorgio Fuà. The team's contributions included estimates of value added by sector at constant (1938) prices, implicit deflators by sector and use, and the creation of a comprehensive series on the capital stock from 1881 onwards (Ercolani 1969; Fuà 1965, 1969; Vitali 1969). Yet, the Fuà team did not attempt to rebuild the core of the work by Istat: the estimates of value added at current prices. It is for this reason that many scholars, troubled by flaws in the original data, remained unconvinced by this revision (henceforth referred to as the Istat-Fuà series).

Anyway, in the absence of any other estimates of Italy's national accounts, the Istat-Fuà series were included in all international collections of historical statistics (i.e., Mitchell 1975) and were used by Thornton (1997) in his study on the link between exports and economic growth for six European countries from the mid-XIX century to the eve of WW1.

Meanwhile, some scholars started to present new revised series for individual sectors and industries. Fenoaltea built new estimates of industrial production for the 1861-1913 period (Fenoaltea 1967, 1972, 1982, 1987, 1998b, 1988c, 2003), while a separate index of industrial production from 1861 to 1980 was put together by Carreras (1983, 1992, 1999).

In the early 1990s Maddison (1991) presented a new estimate of Italy's GDP. Maddison used Istat-Fuà series for agriculture and services and industrial series by Fenoaltea (Fenoaltea 1967, 1982, 1987, 1988b, 1988c), In this fashion Maddison noticeably increased the overall growth rate of Italy's GDP: his series increases between 1861 and 1913 by a factor of 2.5, against just 2.1 for the Istat-Fuà estimates. Nonetheless, this series remained dominated by the Istat components, and apart from a trend correction its path remains extremely close to that of the Istat-Fuà aggregate.

The criticism of the Istat-Fuà series was so widely accepted that an overhaul of the historical accounts was among the projects sponsored by the Bank of Italy in view of its centenary in 1993. The Bank of Italy's project led initially to the re-estimation of aggregate product at current prices in 1911 (Rey 1992). These estimates were retouched almost a decade later, when parallel current-price estimates were compiled for 1891, 1938, and 1951 (Rey 2000).

The Bank of Italy's team aimed also at re-estimating the time series of agricultural, industrial and service production. As this part of the project was not carried through, members of the team published independently each from the other the results of their sectoral value-added estimates.

Thus Fenoaltea presented a preliminary index of industrial value added for the years 1861-1913 at 1911 prices, obtained by adding to the numerous sector-specific series he had compiled over the decades a set of preliminary estimates for the remaining sectors (Fenoaltea 2002a, 2002b, 2003). Soon thereafter Federico published his preliminary estimates of agricultural production from 1860 to 1910, again at 1911 prices (Federico 2003).

Finally, Fenoaltea (2005, 2006) presented new estimates of GDP in Italy from 1861 to 1913 at 1911 prices: the first not to recombine the component series of the original Istat-Fuà estimates. The new GDP series incorporated Federico (2003) series for agriculture, Fenoaltea (2003) series for industry, and a newly derived series for services that extrapolate the Bank of Italy estimates of their value added in 1911 (Rey 2000).

This new Fenoaltea series is now considered as the more accurate estimate of Italy's annual GDP for the years from Unification to the eve of WW1 and has been used in some recent empirical works (Ciccarelli and Fenoaltea 2007; Federico and Vasta 2010). We also use it in this paper.

Instead, the Istat-Fuà estimates of Italy's annual GDP for the years from 1914 to 1939 have never been revisited and are at present the only data available on that period. So we use them in this paper in the version published by Ercolani (1969).

After 1957, Istat updated its current GDP estimates to account for the so-called grey markets, that is, economic activities that were not recorded in official data. However, Istat tied them in with its previous series back only to 1970. Fortunately, Golinelli and Monterastelli (1990) produced a coherent series of national income data from 1951 to 1989, based on the more recent Istat revisions available at that time.

Thus, for the period after 1951, we use the GDP data provided by Golinelli and Monterastelli (1990) for the years from 1951 to 1969, and Istat's more recent estimates for the years from 1970 to 2004. Our analysis stops in 2004 as since 2005 Istat changed the methodology it used to calculate real GDP by passing from fixed-base to chain indexes. Data are in 1990 prices.²

As to the data on exports, for the period from 1863 to 1939 we use the series of new comprehensive statistics of the Bank of Italy that is based on the very detailed official sources of Italy's foreign trade (*Movimento commerciale del Regno d'Italia*). This series constitutes the new official statistics of Italy's foreign trade for the period prior to WW2. This new source calculates real exports by using a specific deflator for export goods which is different from the GDP deflator that is used to calculate real exports in the Istat-Fuà series used by Thornton (1997)³.

For the period from 1951 to 2004 data on exports are taken from OEEC, *Foreign trade series*, for the years 1951-1961, and from United Nations, *UN Comtrade*, for the period from 1962 to 2004. Also in this case a specific deflator for export goods is used to obtain real exports.

Finally, as no reliable data on Italy's exports are available for the 1940-1950 years, this period has been omitted from the analysis. Thus, this paper tests the ELG hypothesis for Italy by focusing on two separate time spans: 1863-1939 and 1951-2004.

² Italy's GDP data at 1990s prices from 1951 to 1996 are published in Di Palma and Carlucci (1997). For the years from 1997 to 2004 we have relied on Istat's official publication *Annuario statistico italiano*.

³ Some preliminary data drawn from these series have been used in Federico and Vasta (2010) and Vasta (2010).

3. The Empirical Model

The export-GDP nexus is a long run relationship whose analysis requires techniques appropriate for estimating long run equilibrium. A statistical test of a long run relationship must take into account the characteristics of time series data. These data should be tested for comovement over time prior to test for causality between them.

Hence, in this section by adopting a time series approach we test if Italian real export and real GDP are cointegrated, then if unidirectional or bidirectional Granger causality exists, in other word if the ELG hypothesis is supported by available data for Italy.

3.1. Granger causality and cointegration

In the case of time series data a test for the direction of causation is suggested by Granger (1969). A variable X improves the prediction of a variable Y, that is X Granger causes Y, if current Y can be predicted better by using past values of X than by not doing so, given that all other past information in the information set is used. Suppose X and Y are linear *covariance stationary* time series⁴. Thus X and Y can be written as follows:

(1)
$$X_{t} = \sum_{i=1}^{m} a_{i} X_{t-i} + \sum_{j=1}^{n} b_{j} Y_{t-j} + \varepsilon_{t}$$

(2) $Y_{t} = \sum_{i=1}^{m} c_{i} Y_{t-i} + \sum_{j=1}^{n} d_{j} X_{t-j} + u_{t}$

where ε_{t} , u_{t} are zero mean and finite covariance matrix random vector. The *causality test* is

- a) X causes Y if H_0 : $d_j = 0, j = 1,...,n$ is rejected
- b) Y causes X if H_0 : $b_i = 0, j = 1,...,n$ is rejected

Bidirectional causality occurs if both (a) and (b) hold. Unidirectional causality from X to Y occurs if (a) holds but (b) does not. In order to test these null hypothesis in (a) and (b), F statistics are calculated for jointly significance of the d_i in equation (1) and for b_j in equation (2).

For the Granger causation test, the hypothesis of covariance stationarity of the time series used is crucial to avoid spurious results. In general, the levels of the time series are not covariance stationary while their first difference are stationary. The growth rate of these variables (ΔX and ΔX) are stationary, while X and Y are not. If these are the statistical properties of the variables, we can only test for Granger causation by using first difference stationary models, that is

(3)
$$\Delta X_{t} = \sum_{i=1}^{m} a_{i} \Delta X_{t-i} + \sum_{j=1}^{n} b_{j} \Delta Y_{t-j} + \varepsilon_{t}$$

(4) $\Delta Y_{t} = \sum_{i=1}^{m} c_{i} \Delta Y_{t-i} + \sum_{j=1}^{n} d_{j} \Delta X_{t-j} + u_{t}$

⁴ Time series are said to be covariance stationary if their moments up to the second order do not depend on time. Hence, for instance the mean must be constant and the shocks affecting stationary series have only temporary effects. These time series are also said I(0). By contrast a series is said to be difference stationary if its first difference is stationary but the series itself is not. A property of difference stationary series is that they do not have necessarily constant means and the variance grows with time without limit, moreover the shocks affecting them are permanent. These series are also said I(1).

However, the exports-GDP nexus is a long run relationship. If this long run nexus exists but we do not include it in the estimation of model (3) and (4) we have mis-specification and 'spurious causality'. Hence, we have to test for Granger causation, to take into account the possible long run relationship among *the levels (values)* of exports and GDP and not only among the short run dynamics of export and GDP growth. Granger type causality tests for a long run relationship are valid if the relevant variables are found to be *cointegrated*, that is they move together so closely over the long run that they share a stochastic (and possibly also deterministic) trend in common. In this latter case as stressed by Granger (1988) there is a presumption for causality to run in at least one direction.

Suppose X is the Italian real exports (*in logs*) and Y is the Italian real GDP (*in logs*). Moreover suppose these series are not covariance stationary, but they are cointegrated co-moving over time. In this case Granger causality test, as in point (a) and (b) stated before, must be performed on the following ECM models:

(5)
$$\Delta X_{t} = \sum_{i=1}^{m} a_{i} \Delta X_{t-i} + \sum_{j=1}^{n} b_{j} \Delta Y_{t-j} + \delta ECT_{t-1} + \varepsilon_{t}$$

(6) $\Delta Y_{t} = \sum_{i=1}^{m} c_{i} \Delta Y_{t-i} + \sum_{j=1}^{n} d_{j} \Delta X_{t-j} + \gamma ECT_{t-1} + u_{t}$

where *ECT* is the error correction term derived by cointegration analysis representing the long run equilibrium among the variables.

To conclude, the causality testing procedure involves three steps. The first step is to test if real exports (in logs) and real GDP (in logs) are not covariance stationary (*integration analysis*). If the variables are not stationary, the second step is to *test for cointegration* using Johansen (1991) multivariate procedure to check for a common trend. As shown by Kremers et al. (1992) this procedure is currently the most reliable test for common trends. Finally, if cointegration exists, then either unidirectional or bidirectional *Granger causality* must exist in at least the stationary variables and causality F-test must be performed on the ECM models above.

4.2. Integration and cointegration analysis

Before testing for causality we check for stationarity by using Dickey Fuller tests (ADF) for the levels and first differences of the variables. We also use tests with stationarity as null (KPSS test) to confirm the results of the usual Dickey Fuller tests of non stationarity. If both reject their nulls then we have no confirmation, but if test ADF rejects the null but test KPSS does not (or viceversa) we have confirmation (see Table 1).

Test ADF (Dickey Fuller)*	Test KPSS**
H_0 : y_t non stationary series (unit root)	H_0 : y_t stationary series (no unit root)
$H1: y_t$ stationary series	$H1: y_t$ non stationary series (unit root)

Table 1 – Tests and confirmation analysis

Notes: * see, Dickey Fuller (1979). ** see, Kwiatkowski, Phillips, Schmidt e Shin (1992).

Table 2 summarises the non stationary results for the period 1863-1939, while Table 3 shows the corresponding results for the period 1951-2004. Both ADF and KPSS depend on a parameter (k or w) which must be chosen in advance depending on the autocorrelation structure of the data: we write ADF(k) and KPSS (w), where k indicates the lags while w the window sizes. We apply both

ADF and KPSS for different values of k and the KPSS for different w. Other details on these tests are in the Notes of the tables.

As shown in Tables 2 and 3, for real exports and real GDP *in levels*, stationarity is rejected with different KPSS window sizes and the non stationarity is not rejected with different lags of the ADF. For the *first differences* of the variables the vice-versa holds. We conclude that real exports and real GDP are non stationary series, while their growth rates are stationary in both samples.

The next step is to test for cointegration to check for a common trend in export and GDP. The Johansen's maximum likelihood method (Johansen 1991), using the Maximum Eingenvalue and Trace statistics, finds no cointegration on the period 1863-1939, while it suggests a significant long run relationship between exports and GDP on the sub-sample 1863-1913, that is excluding the WW1 years and the collapse of international trade in the 1930s following the Great Depression. Cointegration is also the outcome for the period 1951-2004. Such results are robust to varying the length of the model. The statistics of Johansen's cointegration analysis for a VAR with three lags are presented in Table 4, as suggested by information criteria.

If exports and GDP are cointegrated, Granger causality test must be performed on the coefficients of VECM models as described by equations (5) and (6) above. In our case, two ECM models with two lags are estimated and the results of the F-statistics for the conditions (a) and (b) above are presented in Table 5. The F-statistics on the coefficients of GDP growth and export growth are significant. The causal inference between exports and GDP growth is summarized in the last column of the Table. The results suggest that the ELG hypothesis is confirmed only for the period 1951-2004, while in the period 1863-1913 there is evidence for GLE, that is the growth of GDP caused that of exports.

If cointegration is rejected, suggesting no long run comovements among exports and GDP, in the period 1914-1939, it is possible to analyse short run comovements among these variables and test for Granger causation by using the first difference stationary models (3) and (4). However, the turmoil due to WW1 and to the consequences of the Great Depression in the 1930s induced parameter instability and some significant structural breaks in the time series⁵. Because of this parameter instability we do not perform causality tests on the period 1863-1913.

Thus, on the basis of such results we conclude that exports and GDP are generally cointegrated and therefore casually related. In particular, in the years prior to WW1 we find an unidirectional Granger-causality from real GDP to real exports which is just the reverse of the result obtained by Thornton (1997) for the same period. Such a circumstance is due to the fact that we use new and more accurate estimates of both exports and GDP and, above all, to the different methodologies used to estimate real exports. In fact, Thornton (1997) derives not only real GDP but also real exports by using the same GDP deflator, which increases the co-movement of the two aggregates. On the contrary, we derive real exports by using a specific export goods deflator with the consequence of reducing the co-movement of the two variables.

Conversely, for the post-WW2 period we find that the expansion of exports prompted the growth of GDP. This result contrasts with the outcome of the first tests of the ELG hypothesis for Italy by Sharma *et al.* (1991) and Pomponio (1996), but is in line with Riezman *et al.* (1996) trivariate Granger causality test and with Federici and Marconi (2002) test of ELG for the Italian economy from 1960 to 1998.

However, our model cannot contribute to the debate on the timing of the ELG in the years of the Golden Age. In fact, we find no evidence to either support or reject Ciocca, Filosa and Rey (1975) argument that the rapid expansion of exports post-dated the first phase of the GDP high-growth period by almost a decade and the creation of the EEC in 1958 was the real turning point that enabled Italy to set in motion the ELG mechanism. What we can argue is that in the long run Italy's economic growth after WW2 was export-led.

⁵ QLR test for a break at unknown date on models (3) and (4) suggest a significant break in 1917 and in 1920. The outcome for the QLR statistics is: F(4,17) = 5.98 for 1917 and F(4,17) = 4.96 for 1920. In both cases the null of no break is rejected. Critical values in Andrew (2003).

Variable (levels): Real GDP (logs)#					
Non stationarity tests: ADF			Stationarity tests: KPSS		
ADF(0)	-3.29	Not reject the null of non	KPSS(0)	0.24	Reject the null of stationary
		stationary series			series
ADF(1)	-3.80^	Not reject the null of non	KPSS(1)	0.14	Reject the null of stationary
		stationary series			series
ADF(4)	-2.51	Reject the null of non	KPSS(4)	0.09	Not reject the null of
		stationary series			stationary series
		Variable (first differen	ces): ∆Real GI	DP (logs)*	
Non stationa	rity tests: ADI	7	Stationarity	tests: KPSS	
ADF(0)	-8.01	Reject the null of non	KPSS(0)	0.04	Not reject the null of
		stationary series			stationary series
ADF(1)	-6.83	Reject the null of non	KPSS(1)	0.03	Not reject the null of
		stationary series			stationary series
ADF(4)	-4.39	Reject the null of non	KPSS(4)	0.06	Not reject the null of
		stationary series			stationary series
Variable (levels): Real Export (logs)#					
Non stationa	rity tests: ADI	7	Stationarity	tests: KPSS	
ADF(0)	-2.62	Not reject the null of non	KPSS(0)	0.64	Reject the null of stationary
		stationary series			series
ADF(1)	-3.10	Not reject the null of non	KPSS(1)	0.35	Reject the null of stationary
		stationary series			series
ADF(4)	-3.52^	Not reject the null of non	KPSS(4)	0.18	Reject the null of stationary
		stationary series			series
Variable (first differences): ΔExport (logs)*					
Non stationarity tests: ADF			Stationarity tests: KPSS		
ADF(0)	-7.95	Reject the null of non	KPSS(0)	0.05	Not reject the null of
		stationary series			stationary series
ADF(1)	-7.09	Reject the null of non	KPSS(1)	0.05	Not reject the null of
		stationary series			stationary series
ADF(4)	-3.61	Not reject the null of non	KPSS(4)	0.06	Not reject the null of
		stationary series			stationary series

Table 2 Integration analysis, 1863-1939

Notes: # Model with constant and trend included: the 95% critical values for the ADF tests is -3.5 and the 99% is -4.15. The 95% critical values for the KPSS tests is 0.14. * Model with constant included: the 95% critical values for the ADF tests is -2.93 and for the KPSS tests is 0.46. ^ do not reject the null of non stationary series at the 1% significance level. For the ADF tests see Fuller (1976) and for the KPSS tests Kwiatkowski, Phillips, Schmidt e Shin (1992).

Variable (levels): Real GDP (logs), I(1)						
Non stationarity tests: ADF			Stationarity tests: KPSS			
ADF(0)	-2.54	Not reject the null of non	KPSS(0)	1.39	Reject the null of stationary	
		stationary series			series	
ADF(1)	-1.72	Not reject the null of non	KPSS(1)	0.68	Reject the null of stationary	
		stationary series			series	
ADF(4)	-2.58	Not reject the null of non	KPSS(4)	0.30	Reject the null of stationary	
		stationary series			series	
		Variable (first difference	es): ∆Real GDI	P (logs), I(0)		
Non stationa	arity tests: AD	<u>F</u>	Stationarity	tests: KPSS		
ADF(0)	-7.27	Reject the null of non	KPSS(0)	0.12	Not reject the null of	
		stationary series			stationary series	
ADF(1)	-4.95	Reject the null of non	KPSS(1)	0.11	Not reject the null of	
		stationary series			stationary series	
ADF(4)	-4.40	Reject the null of non	KPSS(4)	0.13	Not reject the null of	
		stationary series			stationary series	
Variable (levels): Real Export (logs), I(1)						
Non stationarity tests: ADF			Stationarity tests: KPSS			
ADF(0)	-1.22	Not reject the null of non	KPSS(0)	1.22	Reject the null of stationary	
		stationary series			series	
ADF(1)	-0.45	Not reject the null of non	KPSS(1)	0.63	Reject the null of stationary	
		stationary series			series	
ADF(4)	-2.28	Not reject the null of non	KPSS(4)	0.28	Reject the null of stationary	
		stationary series			series	
Variable (first differences): ΔExport (logs), I(0)						
Non stationarity tests: ADF			Stationarity tests: KPSS			
ADF(0)	-6.75	Reject the null of non	KPSS(0)	0.11	Not reject the null of	
		stationary series			stationary series	
ADF(1)	-5.02	Reject the null of non	KPSS(1)	0.11	Not reject the null of	
		stationary series			stationary series	
ADF(4)	-2.51	Not reject the null of non	KPSS(4)	0.10	Not reject the null of	
		stationary series			stationary series	

Table 3 Integration analysis, 1950-2004

Notes: The 95% critical values for the ADF tests is -3.50 (constant and trend included), Fuller (1976). The 95% critical values for the KPSS tests is 0.14 (trend included), Kwiatkowski, Phillips, Schmidt e Shin (1992).

1863-1939				
Rank	Eigenvalues	Statistics: Trace	Statistics: λ-max	
0	0.12	10.53	10.08	
1	0.005	0.45	0.45	
1863-1913				
Rank	Eigenvalues	Statistics: Trace	Statistics: λ-max	
0	0.33	20.66*	19.37*	
1	0.02	1.29	1.29	
1950 - 2004				
Rank	Eigenvalues	Statistics: Trace	Statistics: λ-max	
0	0.40	29.45*	27.34*	
1	0.04	2.11	2.11	

|--|

Notes: The variables under consideration seem to follow a linear trend then we use a VAR with any restriction on the constant. Critical values for this case in Osterwald-Lenun (1992: Table 1.1*, unrestricted constant). The 95% critical values for Trace are: 17.95 and 8.18, while for λ -max are: 14.90 and 8.18. An asterisk indicates significance at the 5% level. The statistics are from a VAR(3), as suggested by information criteria. However, the results are robust to varying the length of the VAR.

	Export growth (Y) on	GDP growth (X)	GDP growth (Y) on E		
Sample	F test	Results	F test Results		Casual
					inference
1863-1913	F(2, 42) = 2.71* (p-value = 0.07)	Reject H_0	F(2, 42) = 0.35** (pvalue = 0.70)	Fail to reject H_0	GDP causes Export
1950-2004	F(2, 46) = 1.68 ** (p-value = 0.19)	Fail to reject H_0	F(2, 46) = 3.52** (pvalue = 0.037)	Reject H_0	Export causes GDP

Table 5 Granger causality tests, (maximum lag in the VECM =2)

Notes: H_0 is the null that X does not cause Y. ** significant at the 5% level, * significant at the 1% level. The statistics are from a VECM(2), because the cointegration analysis is based on a VAR(3).

5. A historical profile of Italian exports

This section presents a historical profile of Italian exports that helps to interpret the results of the econometric test. At the moment of Unification in 1861 Italy was a mainly agricultural country. In that year agriculture accounted for 64 per cent of the working population and 48 per cent of the nation's GDP (Balcet 1997; Fenoaltea 2005). The school system was underdeveloped and the literacy rate did not exceed 25 per cent. The Italian government in the 1860s and 1870s did not provide any support to industrialization. This entailed a relevant discontinuity with the pre-Unification states, which had tried to foster their ailing industries with tariffs and sometimes incentives, even if with disappointing results. Actually, most states (save the Southern Kingdom of the Two Sicilies) had begun to liberalize trade before Unification. The movement was led by Piedmont which almost totally abolished protection in the 1850s. In 1861 its tariff was extended to the whole country and all the remaining duties were abolished two years later by the trade treaty with France (Federico and Giannetti 1999).

Thus, the first 25 years or so after Unification saw a growth of Italian real exports which more than doubled from 1863 to 1887 (from 538 to 1,161 billion lire at 1911 prices) with an average increase of 3.25 per cent a year (see Figure 1). At that time, primary products were largely dominant and represented about 85 per cent of the total flows. Among them, raw silk stood out with about one third of Italy's total exports (Vasta 2010).

However, Italy's economic growth in the two decades that followed Unification was decidedly inferior than expected. This reinforced the arguments of those who, inside the nation's elite, voiced for a more interventionist policy. So in 1878 the government approved a new tariff which imposed low duties on a limited range of manufactured goods (mainly textiles). The protection was substantially increased in 1887, which was approved jointly with a duty on wheat. The new tariff increased the duties on textiles and imposed a new duty on sugar, pig iron and steel products (Federico and Giannetti 1999).

The 1887 tariff triggered a trade war with France (at that time, Italy's larger trade partner) which lasted ten years. As a consequence, from 1888 to 1890 Italian exports fell by almost a fifth (see Figure 1).

The level of aggregate protection peaked in the early 1890s and then decreased until at least the second decade of the 20th century, because the specific duties were reduced by trade treaties and their *ad valorem* equivalent was lowered by the increase in import prices (Federico and Giannetti 1999). Thus, in 1891 Italian real exports returned to grow and reached a peak of 2,500 billion lire in 1913, with an average growth rate of 4.25 per cent a year over the 1891-1913 period (see Figure 1). These were the years of the first wave of industrialization in Italy which turned out to be buoyant to exports too (Roccas 2003).

The 15 years or so that preceded WW1 also saw the beginning of a change in the composition of Italian exports. In fact, at the eve of the war the share of primary products had fallen to 64 percent and, within them, raw silk had dropped to 18 per cent. Conversely, the share of manufactured goods had risen from 15 per cent in 1861 to 36 per cent in 1913: among them, textiles (silk fabrics and cotton fabrics), machinery and transport equipment stood out (Vasta 2010).

The ratio of real exports to real GDP was just above 5 per cent in 1863. In the following twenty years it showed many ups and downs within a general trend of growth which led it to reach 8.4 per cent in 1883. Then a swing brought it down to 6.5 per cent in 1890. Since 1891 it returned to grow and reached 11 per cent in 1906, a value around which it remained until the outbreak of WW1 (see Figure 2).

These figures suggest that – contrary to the Bonelli-Cafagna view – the size of the exporting sector of the Italian economy was too small and did not grow fast enough to trigger ELG before WW1. Moreover, the leading exporting sectors were those that produced primary goods – especially raw

silk – which were not sophisticated enough to generate technological spillovers that could translate into a higher innovation capacity and more rapid growth of the whole economy.⁶

Fortunately, the constraint on the balance of payments exerted by such a limited capacity to export was eased by some important items which contributed towards offsetting Italy's trade deficit: capital inflows from abroad, tourism and the remittances of Italian emigrants. Earnings form tourism always constituted a positive item in the Italian balance of payments, while only after 1900 were emigrants' remittances substantial enough to balance the negative flow of income from foreign capital investment in government bonds and private enterprise. As a result, the overall balance of payments showed no sign of a structural imbalance which might have curbed economic growth (Zamagni 1993).

Conversely, our evidence seems more in line with Fenoaltea's (1988a, 2006) thesis that links Italy's economic growth to international capital inflows: the decisions of international investors boosted investments in constructions, infrastructures, industry, and public utilities in Italy which induced economic growth. More modern plants and higher production capacity in turn prompted Italy's capacity to export.



Fig. 1. Italian exports 1863-1939 (1911 prices)

⁶ Federico (2005) argues that raw silk might have been a leading sector in the smaller State of Lombary-Veneto prior to the Unification but agrees that after 1861 it was too small to lead the growth of the whole Italian economy.





Similarly to most other countries of the European periphery in the late 19th century, Italy used the capital inflows from the core countries to prompt import-substitution industrialization in capital intensive sectors – in particular steel and heavy mechanics – rather than pursue ELG in sectors in which it could have a competitive advantage in the world market. The fall of prices of the goods that were traded in international markets, together with the stabilization of exchange rates that was made possible by foreign loans were all factors that pushed nations towards protectionism to defend domestic infant industries (De Cecco 2003).

During WW1 exports more than halved and their share of Italy's GDP fell sharply as well since all the resources available in the national economy were mobilised to sustain the war effort (Federico 1998). Once the post-war re-conversion was completed in the early 1920s, exports recovered quickly and reached a peak in 1929 when they accounted for 12 per cent of national output.

However, international trade collapsed in the 1930s as a consequence of the Great Depression and a return to protectionism became a generalized practice all over in the world. Italian exports were severely affected by these events and both their value and share of national GDP fell dramatically in that decade.

The layout of Italy's industrial protection changed somewhat in 1921, when a new tariff raised duties on chemicals, engineering goods, and steel products and lowered those on textiles. The openness of the Italian economy was then remarkably reduced by the autarky policy of the fascist government. This raised tariffs in the late 1920s and imposed quotas in the early 1930s (Federico and Giannetti 1999).

The years between the two world wars saw a further reduction of the share of primary products on Italy's exports which, at the eve of WW2, had decreased to half of the total. For the first time, the share of manufactures reached that of primary products. At the same time, there was also a reduction in exports' concentration. Exports of raw silk collapsed and almost disappeared with a share that in 1939 dropped to two per cent. The top product had become a manufactured one, cotton fabrics, followed by dried fruits and artificial fibres (Vasta 2010).

Fig. 3. Italian exports 1951-2004 (1990 prices)



After WW2 the new democratic government decided trade liberalization. After so many years and the war inflation, the old 1921 tariff was clearly obsolete. Italy substituted it with a new one in 1950, which entailed a return to the moderate protectionism of the 1920s. Behind this shield Italy could afford to abolish all quotas in 1951, well before any other country in the OECD. In the same year, Italy took a decisive step towards liberalization by joining the European Community for Steel and Coal. In spite of the gloomy predictions of many industrialists and experts, the Italian steel industry was not swept away by foreign competition. In the years following, Italy was quite active in the foundation of the EEC in 1957. The liberalization of imports was gradual, and duties were totally abolished in 1968 (Federico 1999; Federico and Giannetti 1999). Moreover, the USA agreed that the exchange rate of the Italian lira within the Bretton Woods international monetary system were established at a value low enough to allow the development of an exporting industry in Italy (De Cecco 2003).

In this context, the post-WW2 years saw an impressive and long-lasting growth of Italian real exports that passed form 9,105 billion lire in 1951 to 456,362 billion lire in 2001 (at 1990 prices), with a staggering average growth rate of 8.14 per cent a year over fifty years (see Figure 3). The exports' annual growth rate was 12.16 per cent from 1951 to 1958. It further rose to 13.76 in the decade that followed the creation of the EEC. Then it declined to 5.55 per cent in the 1970s and to 4.81 per cent in the 1980s, but it rose again to 6.33 per cent in the 1990s.

However, this long phase of growth of Italian exports seems to have come to an end at the beginning of the XXI century and in the years from 2002 to 2004 the 2001 peak was not reached again.

Over the 1951-2001 period exports grew much faster than national output and this had the consequence of progressively enhancing their weight on the GDP. It is worth noticing that in the immediate post-WW2 years the ratio of real exports to real GDP was much lower not only than the late 1920s peak, but also than that of the 1930s⁷. In fact, in 1951 exports accounted only for 4 per

⁷ Of course, our data on the share of exports to GDP are affected by the fact that we calculated real exports by using a specific deflator for export goods which was different from the GDP deflator. If the exports to GDP ratio had been calculated at current prices – as it was commonly the case in the literature (Federico 1998; Roccas 2003) – it would have amounted to about 10 per cent in the early 1950s.

cent of Italy's GDP. The 5 per cent threshold was crossed only in 1957 and the pre-war 12 per cent peak only in 1969. Then the weight of exports on national output further rose in the 1970s (16.3 per cent in 1979), slowed down its growth pace in the 1980s (17.8 per cent in 1989), and soared in the 1990s up to a staggering 29.5 per cent peak in the years 2000 and 2001 (see Figure 4). As a result, the cumulative growth of both real exports and their weight on real GDP over a period of half a century triggered ELG and became the determinant of the growth of the national economy in the long run.



Fig. 4. Ratio of exports to GDP in Italy (1951-2004)

We can also observe that after the wave of strikes and social unrest of the 'hot autumn' of 1969 put an end to a long period of fast and steady growth of Italian exports principally based on the competitive advantage provided by cheap labour costs (Gomellini and Pianta 2007), there were two phases in which the expansion of Italian exports and of their share of GDP grew particularly fast. These were the years 1973-1979 and 1993-1995 that followed the two larger devaluations of the Italian lira.

It is possible to argue that the passage in the early 1970s from fixed to fluctuating exchange rates as a consequence of the collapse of Bretton Woods international monetary system and the possibility to rely on periodical devaluations of the national currency was one of the main drivers that enabled Italy to boost its exports for so long a time and therefore keep ELG functioning in the long run (De Cecco 2000).

However, the beginning of the XXI century seems to have marked a turning point in this respect. In fact, in 2002 Italy abandoned the lira to join the newly-created European single currency, the euro. Such a circumstance prevents now Italy from resorting to competitive devaluations to prompt its exports and is probably a major reason of the reduction of Italian real exports and their share of GDP in the first years of the XXI century.

The post-WW2 years saw also a big change in the structure of Italian exports, that was actually concentrated from the early 1950s and the early 1970s. First of all we can notice a strong reduction in primary products, which passed from more than one third to less than one sixth of the total. Conversely, manufactures rose from 65 per cent in the early 1950s to 84 per cent in the early 1970s. This shift was accompanied by an even bigger change in the composition of manufactured exports: on the one hand there was a decline in traditional products, especially textiles and clothing, while on the other hand there was a strong increase in the share of mechanical products (in particular machinery and transport equipment) – which jumped from 20 per cent to 35 per cent of Italy's total

exports – and, to a lesser extent, of chemicals. In brief, the fast growth of exports was accompanied by a change in their composition, which became more similar to that of more industrialized nations. By contrast, in the period since the mid-1970s the growth of Italian exports was accompanied by only minor changes in their composition: primary products went further down to 10 per cent of the total while manufactures went slightly up to 90 per cent. Among the latter, mechanical goods rose to 37 per cent. Thus, in the last thirty years Italian exports became polarized in two categories, the first one pertaining to the traditional sectors of the 'Made in Italy' (which includes personal and household goods such as textiles, clothing, leather, footwear, wood, tiles, furniture, jewellery, cosmetics, musical instruments, toys and sports items), and the second one to mechanical productions largely constituted by the machinery to manufacture the former (Vasta 2010). It is someway possible to argue that the ELG of the 'Made in Italy' sector led the growth of the machinery sector that at a certain point in time also became export-led. As the latter was a capital

machinery sector that at a certain point in time also became export-led. As the latter was a capital goods industry it could more easily generate those technological spillovers that spread throughout the economy and translated into a higher overall economic growth.

6. Conclusions

This paper has investigated the causal relationship between real exports and real GDP in Italy from 1863 to 2004 by using cointegration analysis and causality tests. The outcome suggests that in the period prior to WW1 the growth of the Italian economy led that of exports, while in the post-WW2 period the causal relationship was reversed with the expansion of exports that determined that of the GDP. For the years from 1914 and 1939 we found no long run relationship between export and economic growth as a consequence of the fall of Italian exports during WW1 and of the collapse of world trade in the years that followed the Great Depression.

This paper does not support the Bonelli-Cafagna view that the growth of the Italian economy in the years from Unification (or earlier) to WW1 was led by a long wave of growth and accumulation stimulated by an expansion of exports, particularly of agricultural goods and raw silk. We argue instead that in the fifty years prior to WW1 the weight of the exporting sector on the Italian economy was too small and did not grow fast enough to trigger ELG. Moreover, the exports consisted largely of primary products that could not generate technological spillovers capable of translating into a higher rate of innovation and more rapid growth of the whole economy. Conversely, our results are consistent with Fenoaltea (1988a, 2006) thesis that links Italy's economic growth to international capital inflows: these boosted investments in constructions, infrastructures, industry, and public utilities which induced economic growth. More modern plants and higher production capacity in turn prompted Italy's capacity to export.

The 1951-2004 years saw an impressive and long-lasting growth of Italian real exports and on their weight on national output which enabled to set in motion ELG. The opening-up of European markets as a consequence of the creation of the EEC and, since the 1970s, the possibility of resorting to recursive devaluations of the Italian lira were among the major factors that fostered the expansion of Italian exports in that period. Italian exports became increasingly specialized in the durables of the 'Made in Italy' and in the machinery to make them. As the latter was a capital goods industry it could more easily generate those technological spillovers that spread throughout the economy and translated in a higher overall economic growth.

Lastly, our results are coherent with the 'threshold effect' thesis. It is possible to argue that in the 1863-1913 period Italy was still a mainly agricultural country that had not achieved a stage of development – or level of industrialization – high enough to engender ELG. Conversely, the situation had reversed in the 1951-2004 years. The manufacturing sector accounted for a much larger share of the economy than before WW1 and was now able to trigger ELG.

REFERENCES

Andrew, D.A., (2003), 'Test for parameter instability and structural change with unknown change point: a corrigendum', *Econometrica*, 71, 395-397.

Awokuse, T.O. (2003), 'Is the export-led growth hypothesis valid for Canada?, *Canadian Journal of Economics*, 36, 126-136.

Awokuse, T.O. (2006), 'Export-led growth and the Japanese economy: evidence from VAR and directed acyclic graphs, *Applied Economics*, 38, 593-602.

Balassa, B. (1978), 'Exports and economic growth: further evidence', Journal of Development Economics, 5, 181-189.

Balassa, B. (1984), 'Adjustment to external shocks in developing economies', World Bank Staff Working Paper 472 (Washington: The World Bank).

Balcet, G. (1997), L'economia italiana. Evoluzione, problemi e paradossi (Milan: Feltrinelli).

Battilani, P. and Fauri, F. (2008), Mezzo secolo di economia italiana. 1945-2008 (Bologna: Il Mulino).

Bonelli, F. (1978), 'Il capitalismo italiano: linee generali di interpretazione', in R. Romano and C. Vivanti (eds), *Storia d'Italia. Annali I. Dal feudalesimo al capitalismo* (Turin: Einaudi), pp. 1193-1255.

Burney, N.A. (1996), 'Exports and economic growth: evidence from cross-country analysis', *Applied Economic Letters*, 3, 369-373.

Cafagna, L. (1989), Dualismo e sviluppo nella storia d'Italia (Venice: Marsilio).

Carreras, A. (1983), *La producció industrial espanyola i italiana des mitjan segle XIX fins a l'actualitat*, Unpublished Ph.D dissertation, Universitat Autonoma de Barcelona.

Carreras, A. (1992), 'La producción industrial en el muy largo plazo: una comparación entre España e Italia de 1861 a 1980', in L. Prados and V. Zamagni (eds), *El desarrollo económico en la Europa del Sur: España e Italia en perspectiva histórica* (Madrid: Alianza), pp. 173-210.

Carreras, A. (1999), 'Un ritratto quantitativo dell'industria italiana', in F. Amatori, D. Bigazzi, R. Giannetti, L. Segreto (eds), *Storia d'Italia. Annali 15, L'industria* (Turin: Einaudi), pp. 179-272.

Chenery, H.B. and Strout, A. (1966), 'Foreign assistance and economic development', American Economic Review, 679-732.

Ciccarelli and Fenoaltea, S. (2007), 'Business Fluctuations in Italy, 1861-1913: The new evidence', *Explorations in Economic History*, 44, 432-451.

Ciocca, P., Filosa, R. and Rey G. (1975), 'Integration and Development of the Italian Economy, 1951-1971: A Re-Examination', *Banca Nazionale del Lavoro Quarterly Review*, 29, 284-320.

Cohen, J. and Federico, G. (2001), *The Growth of the Italian Economy 1820-1960* (Cambridge: Cambridge University Press).

Crivellini, M. (1993), Aspetti macroeconomici dello sviluppo dell'economia italiana, Università di Ancona, Dipartimento di Economia, Studi sullo sviluppo, n. 3.

Dar, A., and Amirkhalkhali, S. (2003), 'On the impact of trade openness on growth: further evidence form OECD countries', *Applied Economics*, 35, 1761-1766.

De Cecco, M. (2000), L'economia di Lucignolo: opportunità e vincoli dello sviluppo italiano (Rome: Donzelli).

De Cecco, M. (2003), 'L'Italia grande potenza: la realtà del mito', in P. Ciocca and G. Toniolo (eds), *Storia economica d'Italia*, vol. 3: *Industrie, mercati, istituzioni*, t. 2: *I vincoli e le opportunità* (Rome-Bari: Laterza), pp. 3-36.

De Nardis, S. and Traù, F. (2005), *Il modello che non c'era. L'Italia e la divisione internazionale del lavoro industriale* (Soveria Mannelli: Rubbettino).

Di Palma, M. and Carlucci, M. (1997), 'L'evoluzione dei principali aggregati economici nell'ultimo cinquantennio', in M. Arcelli (ed.), *Storia, economia e società in Italia 1947-1997* (Rome-Bari: Laterza), pp. 351-84.

Dickey, D.A. and Fuller, W.A. (1979), 'Distributions of the estimators for autoregressive time series with a unit root', *Journal of the American Statistical Association*, 74, 427-81.

Ercolani, P. (1969), 'Documentazione statistica di base', in G. Fuà (ed.), *Lo sviluppo economico in Italia*, vol. 3 (Milan: Franco Angeli).

Esfahani, S.H. (1991), 'Exports, imports and economic growth in semi-industrial countries', *Journal of Development Economics*, 35, 93-116.

Federici, D. and Marconi, D. (2002), 'On exports and economic growth: the case of Italy', *Journal of International Trade & Economic Development*, 11(3), 323-340.

Federico, G. (1996), 'Italy, 1860-1940: a little-known success story', Economic History Review, 49(4), 764-786.

Federico, G. (1998), 'Commercio' in M. Firpo, N. Tranfaglia and P.G. Zumino (eds), *Guida all'Italia contemporanea*, vol. 1: *Risorse e strutture economiche* (Milan: Garzanti), pp. 425-70.

Federico, G. (1999), 'Harmful or Irrelevant? Italian Industrial Policy, 1945-1973', in H. Miyajima, T. Kikkawa and T. Hikino (eds), *Policies for Competitiveness. Comparing Business-Government Relationships in the 'Golden Age of Capitalism'* (Oxford: Oxford University Press), pp. 309-335.

Federico, G. (2003), 'Le nuove stime della produzione agricola italiana, 1860-1910: primi risultati e implicazioni', *Rivista di storia economica*, 19, 357-81.

Federico, G. (2005), 'Seta, agricoltura e sviluppo economico in Italia', Rivista di storia economica, 21(2), 123-54.

Federico, G. and Giannetti, R. (1999), 'Italy: Stalling and Surpassing', in J. Foreman-Peck and G. Federico (eds), *European Industrial Policy. The Twentieth-Century Experience* (Oxford: Oxford University Press), pp. 124-151.

Federico, G., Tattara, G. and Vasta, M. (forthcoming), *Le statistiche del commercio estero italiano, 1863-1939* (Rome: Banca d'Italia).

Federico, G. and Vasta, M. (2010), 'Was industrialization an escape from commodity lottery? Evidence from Italy, 1861-1939', *Explorations in Economic History*, 47(2), 228-43.

Fenoaltea, S. (1967), *Public Policy and Italian industrial development, 1861-1913*, Unpublished Ph.D dissertation, Harvard University.

Fenoaltea, S. (1972), 'Railroads and Italian industrial growth, 1861-1913', *Explorations in Economic History*, 9, 325-51.

Fenoaltea, S. (1973), 'Riflessioni sull'esperienza industriale italiana dal Risorgimento alla prima guerra mondiale', in G. Toniolo (ed.), *Lo sviluppo economico italiano 1861-1940* (Bari: Laterza), 121-156.

Fenoaltea, S. (1982), 'The growth of the utilities industries in Italy, 1861-1913', *Journal of Economic History*, 42, 601-27.

Fenoaltea, S. (1987), 'Construction in Italy, 1861-1913', Rivista di storia economica, 4, 21-53.

Fenoaltea, S. (1988a), 'International Resource Flows and Construction Movements in the Atlantic Economy: The Kuznets Cycle in Italy, 1861-1913', *Journal of Economic History*, 48(3), 605-637.

Fenoaltea, S. (1988b), The extractive industries in Italy, 1861-1913: general methods and specific estimates', *Journal of European Economic History*, 17, 117-25.

Fenoaltea, S. (1988c), 'The growth of Italy's silk industry, 1861-1913: a statistical reconstruction', *Rivista di storia economica*, 5, pp. 605-38.

Fenoaltea, S. (2002a), 'Production and consumption in post-unification Italy: new evidence, new conjectures', *Rivista di storia economica*, 18, 251-98.

Fenoaltea, S. (2002b), 'Lo sviluppo dell'industria dall'Unità alla Grande Guerra: una sintesi provvisoria', in P. Ciocca and G. Toniolo (eds.), *Storia economica d'Italia*, vol. 3.1: *Le strutture dell'economia*, Roma-Bari, Laterza.

Fenoaltea, S. (2003), 'Notes on the rate of industrial growth in Italy, 1861-1913', *Journal of Economic History*, 63, 695-735.

Fenoaltea, S. (2005), 'The growth of the Italian economy, 1861-1913: Preliminary second-generation estimates', *European Review of Economic History*, 9, 273-312.

Fenoaltea, S. (2006), L'economia italiana dall'unità alla grande guerra (Rome-Bari: Laterza).

Fuà, G. (1965), Notes on Italian Economic Growth 1861-1964 (Milan: Giuffrè).

Fuà, G., ed. (1969), Lo sviluppo economico in Italia, 3 vols (Milan: Franco Angeli).

Fuller (1976), Introduction to Statistical Time Series (New York: Wiley).

Gallino, L. (2003), La scomparsa dell'Italia industriale (Turin: Einaudi).

Gerschenkron, A. (1962), *Economic Backwardness in Historical Perspective* (Cambridge, MA.: Cambridge University Press).

Geweke, J. (1984), 'Measures of conditional linear dependence and feedback between time series', *Journal of the American Statistical Association*, 79(388), 907-915.

Ghartey, E.E. (1993), Causal Relationship between exports and economic growth – some empirical evidence in Taiwan, Japan and the United States', *Applied Economics*, 25, 1145-1152.

Giles, J.A. and Williams, C.L. (2000a), 'Export-led Growth: A Survey of the Empirical Literature and Some Noncausality Results. Part 2., *Journal of international trade and economic development*, 9, 265-341.

Giles, J.A. and Williams, C.L. (2000b), 'Export-led Growth: A Survey of the Empirical Literature and Some Noncausality Results. Part 2., *Journal of international trade and economic development*, 9(4), 445-470.

Golinelli, R. and Monterastelli, M. (1990), Un metodo per la ricostruzione di serie storiche compatibili con la nuova contabilità nazionale, 1951-1989, Prometeia Report No. 9001, Bologna.

Gomellini, M. and Pianta, M. (2007), 'Commercio con l'estero e tecnologia in Italia negli anni cinquanta e sessanta', in AA.VV., *Innovazione tecnologica e sviluppo industriale nel secondo dopoguerra* (Rome-Bari: Laterza), 359-594.

Granger, C.W.J. (1969), 'Investigating causal relations by econometric models: cross spectral methods', *Econometrica*, 37, 424-38.

Granger, C.W.J., 'Some recent developments in the concept of causality', Journal of Econometrics, 39, 199-211.

Graziani, A., ed. (1969), Lo sviluppo di un'economia aperta (Naples: Esi).

Graziani, A. (1998), Lo sviluppo dell'economia italiana. Dalla ricostruzione alla moneta europea. Nuova edizione aggiornata (Turin: Bollati Boringhieri).

Greenaway, D., and Sapsford, D. (1994), 'Exports, Growth, and Liberalization: An Evaluation', *Journal of Policy Modeling*, 16(2), 165-186.

Grossman, G.M. and Helpman, E. (1991), Innovation and Growth in the Global Economy (Cambridge, MA: MIT Press)

Helpman, E. and Krugman, P.R. (1985), Market Structure and Foreign Trade (Cambridge, Mass.: MIT Press).

Istat (1957), 'Indagine statistica sullo sviluppo del reddito nazionale dell'Italia dal 1861 al 1956', Annali di statistica, Serie VIII, 9.

Johansen, S. (1991), 'Estimation and hypothesis test of cointegrating vectors in Gaussian vector autoregressive models', *Econometrica*, 59, 1551-80.

Kaldor, N. (1970), 'The case for regional policies', Scottish Journal of Political Economy, 337-348.

Kavoussi, R.M. (1984), 'Export Expansion and Economic Growth', Journal of Development Economics, 14, 241-250.

Keesing, D.B. (1967), 'Outward-looking policies and economic development', Economic Journal, 77, 303-20.

Kremers et al. (1992), The power of co-integration tests, Oxford Bulletin of Economics and Statistics, 54, 325-48.

Krueger, A.O. (1978), Foreign Trade Regimes and Economic Development: Liberalization Attempts and Consequences (Cambridge, MA: Balinger).

Krugman, P.R. (1984), 'Import Protection as Export Promotion', in H. Kierzkowski (ed.), *Monopolistic Competition in International Trade* (Oxford: Oxford University Press).

Kwiatkowski, D., Phillips, P.C.B., Schmidt, P. and Shin, Y. (1992), 'Testing the null hypothesis of stationarity against the alternative of a unit root', *Journal of Econometrics*, 54, 159-78.

Lancaster, K. (1980), 'Intra-industry trade under perfect monopolistic competition', *Journal of International Economics*, 10, 151-175.

Lanza, A. and Quinteri, B. (2007), 'Quote di mercato e qualità delle esportazioni italiane: il quadro generale', in A. Lanza and B. Quinteri (eds), *Eppur si muove. Come cambia l'export italiano* (Soveria Mannelli: Rubbettino), pp. 11-33.

Maddison, A. (1991), 'A revised estimate of Italian eocnomic growth, 1861-1989, BNL Quarterly Review, 177, 225-41.

McKinnon, R. (1964), 'Foreign exchange constraint in economic development and efficient aid allocation', *Economic Journal*, 74, 388-409.

Michaely, M. (1977), 'Exports and Growth: An Empirical Investigation', Journal of Development Economics, 4, 49-53.

Mitchell, B.R. (1975), European Historical Statistics (London: Macmillan).

Moschos, D. (1989), 'Export Expansion, Growth and the Level of Economic Development', *Journal of Development Economics*, 16, 93-102.

Nurkse, R. (1961), 'Trade theory and development policy', in H.S. Ellis (ed.), *Economic development of Latin America* (New York: St. Martin Press), 236-46.

Onida, F. (2004), Se il piccolo non cresce (Bologna: Il Mulino).

Pack, H. (1988), 'Industrialization and Trade', in H. Chenery and T.N. Srinivasan (eds), *Handbook of Development Economics*, Vol. 1 (Amsterdam: Elsevier).

Pomponio, X.Z. (1996), 'A Causality Analysis of Growth and Export Performance', Atlantic Economic Journal, 24, 168-176.

Ram, R. (1987), 'Exports and Economic Growth in Developing Countries: Evidence from Time-Series and Cross-Section Data', *Economic Development and cultural Change*, 36(1), 51-72.

Rey, G.M., ed. (1992), I conti economici dell'Italia, vol. 2: Una stima del valore aggiunto per il 1911 (Rome-Bari: Laterza).

Rey, G.M., ed. (2000), I conti economici dell'Italia, vol. 3.2: Il valore aggiunto per gli anni 1891, 1938, 1951 (Rome-Bari: Laterza).

Riezman, R. et al. (1996), 'The engine of growth or its handmaiden? A time-series assessment of export-led growth', *Empirical Economics*, 21, 77-110.

Roccas, M. (2003), 'Le esportazioni nell'economia italiana', in P. Ciocca and G. Toniolo (eds), *Storia economica d'Italia*, vol. 3: *Industrie, mercati, istituzioni*, t. 2: *I vincoli e le opportunità* (Rome-Bari: Laterza), pp. 37-135.

Sharma, S.C. et al. (1991), 'Exports and economic growth in industrialized countries', Applied Economics, 23, 697-708.

Stern, R.M. (1967), Foreign Trade and Economic Growth in Italy (New York: Praeger Publishers).

Summers, R. and Hesston, A. (1991), 'The penn-world table (Mark 5): An expanded set of international comparisons, 1950-1988, *Quarterly Journal of Economics*, 106, 327-368.

Thornton, J. (1997), 'Exports and economic growth: Evidence from 19th Century Europe', *Economic Letters*, 55, 235-240.

Tyler, W. (1981), 'Growth and Export Expansion in Developing Countries: Some Empirical Evidence', *Journal of Development Economics*, 3, 337-349.

Vasta, M. (1999), Innovazione tecnologica e capitale umano in Italia (1880-1914). Le traiettorie tecnologiche della seconda rivoluzione industriale (Bologna: Il Mulino).

Vasta, M. (2010), 'Italian export capacity in a long run perspective (1861-2009): a tortuous path to stay in place', *Journal of Modern Italian Studies*, 15(1), 133-156.

Verdoorn, P.J. (1949), I fattori che regolano lo sviluppo della produttivita' del lavoro, L' Industria, 1, 3-11.

Visco, V. et al. (2004), Il declino economico dell'Italia. Cause e rimedi (Milan: Bruno Mondadori).

Vitali, O. (1969), 'La stima del valore aggiunto a prezzi costanti per rami di attività', in G. Fuà (ed.), *Lo sviluppo economico in Italia*, vol. 3 (Milan: Franco Angeli).

Yaghmaian, B., and Ghorashi, R. (1995), 'Export Performance and Economic Development: An Empirical Analysis', *The American Economist*, 39(2), 37-45.

Zamagni, V. (1992), 'The Italian "Economic Miracle" Revisited: New Markets and American Technology', in E. Di Nolfo (ed.), *Power in Europe*, Vol. 2: *Great Britain, France, Germany and Italy and the Origins of the EEC. 1952-1957* (Berlin: De Gruyter), pp. 197-226.

Zamagni, V. (1993), The Economic History of Italy 1860-1990. Recovery after Decline (Oxford: Clarendon Press).

"Materiali di Discussione" LATER PUBLISHED ELSEWHERE

- N. 546 M. Murat and B. Pistoresi, *Emigrants and immigrants networks in FDI*, Applied Economics letters, April 2008, <u>http://www.informaworld.com/</u> /<u>content~content=a789737803~db=all~order=author</u> (electronic publication), WP No. 546 (December 2006).
- N. 545 M.Brunetti and C. Torricelli, *The Population Ageing in Italy:Facts and Impact on Household Portfolios*, in M. Balling & E. Gnan & F. Lierman (eds.), Money, Finance and Demography: The Consequences of Ageing, Vienna, Suerf (2007), WP No. 545 (November 2006).
- N. 532 M: Montanari, Between European Integration and Regional Autonomy: The Case of Italy from an Economic Perspective, Constitutional Political Economy, Vol. 17, 4, pp. 277-301 (2006), WP No. 532 (March 2006).
- N. 529 M. Montanari, *Knocking on the EU's door: the Political Economy of EU-Ukraine Relations*, Journal of Contemporary European Research, Vol.3,1, pp.64-78 (2007), **WP No. 529 (February 2006).**
- N. 518 M.Brunetti and C. Torricelli, *Economic Activity and Recession Probabilities: information content and predictive power of the term spread in Italy*, Applied Economics (2009), **WP No. 518 (December 2005).**
- N. 517 M. Murat and S. Paba (2006), *I distretti industriali tra immigrazioni e internazionalizzazione produttiva*, in B. Quintieri (ed.) I distretti italiani dal locale al globale, Rubbettino (2006), WP No. 517 (December 2005).
- N. 491 V. Moriggia, S. Muzzioli and C. Torricelli, On the no arbitrage condition in option implied trees, European Journal of Operational Research (2009), WP No. 491 (May 2005).
- N. 482 G. Di Lorenzo and G. Marotta, A less effective monetary transmission in the wake of EMU? Evidence from lending rates passthrough, ICFAI Journal of Monetary Economics, Vol. 4, 2, pp. 6-31 (2006), WP No. 482 (February 2005).
- N. 472 M.Brunetti and C. Torricelli, *The internal and cross market efficiency in index option markets: an investigation of the Italian market*, Applied Financial Economics, Vol. 17, 1, pp. 25-33 (2007), WP No. 472 (November 2004).
- N. 466 G. Marotta, *La finanza del settore non profit tra ritardi nei pagamenti e Basilea 2*, Banca Impresa Società , Vol. XXIV, 1, pp. 35-51 (2005), WP
 No. 466 (September 2004).

- N. 453 Pederzoli and C. Torricelli, Capital requirements and Business Cycle Regimes: Forward-looking modelling of Default Probabilities, Journal of Banking and Finance, Vl. 29, 12, pp. 3121-3140 (2005), WP No. 453 (February 2004).
- N. 448 V. Moriggia, S. Muzzioli, C. Torricelli, *Call and put implied volatilities and the derivation of option implied trees,* Frontiers In Finance and Economics, vol.4, 1, pp. 35-64 (2007), **WP No. 448 (November 2003).**
- N. 436 M.Brunetti and C. Torricelli, *Put-Call Parity and cross-market efficiency* in the Index Options Markets: evidence from the Italian market, International Review of Financial Analysis, VI.14, 5, pp. 508-532 (2005), WP No. 436 (July 2003).
- N. 429 G. Marotta, When do trade credit discounts matter? Evidence from Italian Firm-Level Data, Applied Economics, Vol. 37, 4, pp. 403-416 (2005), WP No. 429 (February 2003).
- N. 426 A. Rinaldi and M. Vasta, *The Structure of Italian Capitalism*, 1952-1972: New Evidence Using the Interlocking Directorates Technique, Financial History Review, vol, 12, 2, pp. 173-198 (2005), WP No. 426 (January 2003).
- N. 417 A. Rinaldi, *The Emilian Model Revisited: Twenty Years After*, Business History, vol. 47, 2, pp. 244-226 (2005), **WP No. 417 (September 2002).**
- N. 375 G. Marotta, La direttiva comunitaria contro i ritardi nei pagamenti tra imprese. Alcune riflessioni sul caso italiano, Banca, Impresa, Società, Vol. XX, 3, pp. 451-71 (2001), WP No. 375 (September 2001).
- N. 303 G. Marotta and M. Mazzoli, *Fattori di mutamento nella domanda di prestiti ed effetti sulla trasmissione della politica monetaria*, in P. ALESSANDRINI (ed.) Il sistema finanziario italiano tra globalizzazione e localismo, Bologna, Il Mulino, pp. 223-260 (2001), WP No. 303 (April 2000)
- N. 131 G. Marotta, *Does trade credit redistribution thwart monetary policy? Evidence from Italy*, Applied Economics, Vol. 29, December, pp. 1619-29 (1997), **WP No. 131 (1996).**
- N. 121 G. Marotta, *Il credito commerciale in Italia: una nota su alcuni aspetti strutturali e sulle implicazioni di politica monetaria*, L'Industria, Vol. XVIII, 1, pp. 193-210 (1997), **WP No. 121 (1995)**
- N. 105 G. Marotta, *Credito commerciale e "lending view"*, Giornale degli Economisti e Annali di Economia, Vol. LIV, 1-3, gennaio-marzo, pp. 79-102; anche in G. Vaciago (a cura di) *Moneta e finanza*, Bologna, Il Mulino (1995), WP No. 105 (1994).

RECENTLY PUBLISHED "Materiali di Discussione"

N. 654 -	<i>Transnational social capital and FDI. Evidence from Italian associations worldwide</i> , by Marina Murat, Barbara Pistoresi and Alberto Rinaldi [May 2011].
N. 653 -	Le difficoltà d'accesso all'abitazione da parte degli immigrati. Una indagine di campo, by Paola Bertolini, Francesco Pagliacci and Serena Giannuzzi [April 2011].
N. 652 -	Il social housing in Europa, by Massimo Baldini Baldini and Marta Federici [April 2011].
N. 651 -	L'edilizia sociale di fronte all'immigrazione a Modena: politiche abitative e domanda potenziale (o inevasa), by Marta Federici and Giuseppe Fiorani [April 2011].
N. 650 -	Dynamic Adverse Selection and the Size of the Informed Side of the Market, by Ennio Bilancini and Leonardo Boncinelli [March 2011].
N. 649 -	<i>Present and Future of the Chinese labour Market</i> , by Michele Bruni and Claudio Tabacchi [March 2011].
N. 648 -	Lisbon strategy and EU countries' performance: social inclusion and sustainability, by Paola Bertolini and Francesco Pagliacci [March 2011].
N. 647 -	Schools choices of foreign youth in Italian territorial areas, by Paola Bertolini, Valentina Toscano and Linda Tosarelli [March 2011].
N. 646 -	Allocation of Time within Italian Couples: Exploring the Role of Institutional Factors and their Effects on Household's Wellbeing, by Tindara Addabbo Antonella Caiumi and Anna Maccagnan [February 2011.
N. 645 -	Measuring the interaction between parents and children in Italian families: a structural equation approach, by Anna Maccagnan [February 2011]
N. 644 -	<i>The Italian Labour Market and the crisis</i> , by Tindara Addabbo and Anna Maccagnan [February 2011].
N. 643 -	China's New Demographic Challenge: From Unlimited Supply of Labour to Structural Lack of Labour Supply. Labour market and demographic scenarios: 2008-2048, by Michele Bruni [February 2011].
N. 642 -	<i>Cardinality versus q-Norm Constraints for Index Tracking</i> , by Bjöern Fastrich, Sandra Paterlini and Peter Winker [January 2011].
N. 641 -	A <i>Multi-Scalar Analysis of European Cities</i> , by Enrico Giovannetti and Francesco Pagliacci [December 2010]
N. 640 -	The Attractiveness of countries for FDI. A Fuzzy Approach, by Marina Murat and Tommaso Pirotti [December 2010].
N. 639 -	Income distribution and the effect of the financial crisis on the Italian and Spanish labour markets, by Tindara AddabboTindara Addabbo, Rosa García-Fernández, Carmen Llorca-Rodríguez and Anna Maccagnan [Dicembre 2010].
N. 638 -	Exaptation, Degeneracy and Innovation, by Giovanni Bonifati [November 2010].